

Author's response to reviews

Title: Tobacco cessation clinical practice guidelines in rural and urban hospitals:
A pre-implementation nursing needs assessment

Authors:

Patricia M Smith (psmith@nosm.ca)
Scott M Sellick (sellicks@tbh.net)

Version: 4 **Date:** 6 May 2011

Author's response to reviews:

The only change we have made is to include the abstract in the manuscript document and re-upload the manuscript.

Thank you.